**Author’s response to reviews**

**Title:** Inhibition of biofilm formation in Mycobacterium smegmatis by Parinari curatellifolia leaf extracts

**Authors:**

Benjamin Bhunu (bhunubenjamin@gmail.com)
Ruvimbo Mautsa (ruemautsa@gmail.com)
Simbarashe Sithole (mandlasithole@gmail.com)
Stanley Mukanganyama (smukanganyama@medic.uz.ac.zw)

**Version:** 3  **Date:** 16 May 2017

**Author’s response to reviews:**

Reviewer reports:

Dinesh Chellappan, Ph.D (Reviewer 3): All suggestions were considered and were modified in the manuscript accordingly by the authors.

Few errors need to be corrected, which are mentioned below.

**Keywords** - The word 'Keywords' is one word. This needs to be rectified. The authors have used the appropriate keyword style in their 'response to the reviewers'. However, this is not reflected in the main manuscript.

The word ‘Key words’ has been corrected to ‘Keywords’. The appropriate keyword style has been reflected in the main manuscript.

**References** - In Reference number 4; there needs to be a standard space between the number and the text.

Reference 4 has been corrected and a standard space between the number and the text and been put. Reference 4 now appears as follows: